Author's response to reviews

Title: Social Network Properties and Self-Rated Health in Later Life Comparisons from the Korean Social Life, Health, and Aging Project and the National Social Life, Health and Aging Project

Authors:

Yoosik Youm (yoosik@yonsei.ac.kr)
Edward O Laumann (ob01@uchicago.edu)
Kenneth F Ferraro (ferraro@purdue.edu)
Linda J Waite (l-waite@uchicago.edu)
Hyeon Chang Kim (hckim@yuhs.ac)
Yeong-Ran Park (yeongranpark@daum.net)
Sang Hui Chu (shchu@yuhs.ac)
Won-tak Joo (yochicken@naver.com)
Jina Lee (lovejirosang@gmail.com)

Version: 8 Date: 28 August 2014

Author's response to reviews: see over
Dear editorial board of BMC geriatrics:

We really appreciate your precious and productive comments. Please find enclosed copy edited version of the manuscript: “Social Network Properties and Self-Rated Health in Later Life: Comparisons from the Korean Social Life, Health, and Aging Project and the National Social Life, Health and Aging Project”, by Yoosik Youm and others.

In the manuscript, we have reflected the editorial comments that were received on August 18th 2014 as follows.

“Yes, the authors should copyedit the manuscript themselves (some grammatical errors are still found, e.g., line 264, "list" instead of "lists"; line 583m remove "a")"

Since Yonsei University, where the corresponding author is affiliated, provides free proofreading and copyediting services, we have requested the university writing center to copyedit the manuscript.

Again, we appreciate your valuable comments and please let us know if you have any other concerns or suggestions.

Sincerely yours,
Yoosik Youm on behalf of the authors of the manuscript.

Corresponding author: Yoosik Youm. Yonsei-ro 50. Sociology Department. Yonsei University, Seoul, South Korea. 120-749., yoosik@yonsei.ac.kr, phone number: 82-2-2123-2431, fax number 82-2-2123-2420